Author's response to reviews

Title: A Follow-Up Study for Left Ventricular Mass on Chromosome 12p11 Identifies Potential Candidate Genes

Authors:

David Della-Morte (ddellamorte@med.miami.edu)
Ashley Beecham (ABeecham@med.miami.edu)
Tatjana Rundek (Trundeked@med.miami.edu)
Liyong Wang (lwang1@med.miami.edu)
Mark S McClendon (MMcClendon2@med.miami.edu)
Susan Slifer (SSlifer@med.miami.edu)
Susan H Blanton (SBlanton@med.miami.edu)
Marco R Di Tullio (md42@columbia.edu)
Ralph L Sacco (rsacco@med.miami.edu)

Version: 6 Date: 25 July 2011

Author’s response to reviews: see over
July 25, 2011

Dr. Jigisha Patel, MRCP, PhD
Series Editors - Medicine
BMC Medical Genetics

Dear Dr. Patel,

We formatted our accepted manuscript (No. MS: 1860612615208250) entitled “A Follow-Up Study for Left Ventricular Mass on Chromosome 12p11 Identifies Potential Candidate Genes” as per BMC Medical Genetics editorial request.

The authors are: David Della-Morte, Ashley Beecham, Tatjana Rundek, Liyong Wang, Mark S. McClendon, Susan Slifer, Susan H. Blanton, Marco R. Di Tullio, and Ralph L. Sacco.

Thank you for accepting our manuscript for publication.
Sincerely yours,
Ralph L. Sacco, MD, MS